

Available online at www.CivileJournal.org

Civil Engineering Journal

(E-ISSN: 2476-3055; ISSN: 2676-6957)

Vol. 10, No. 04, April, 2024



Seismic Resilience of Steel-Braced Frames Incorporating Steel Slit Dampers: A Review and Comparative Numerical Analysis

Zaid A. Al-Sadoon ^{1*}⁽⁶⁾, M. Almohammad-albakkar ²⁽⁶⁾

¹Department of Civil and Environmental Engineering, College of Engineering, University of Sharjah, P.O. Box. 27272, United Arab Emirates. ²Department of Civil Engineering, Isfahan University of Technology, Isfahan 84156-83111, Iran.

Received 08 December 2023; Revised 03 March 2024; Accepted 12 March 2024; Published 01 April 2024

Abstract

Steel dampers, specifically steel slit dampers (SSDs), are crucial for enhancing the seismic resilience of buildings by absorbing energy and mitigating damage. SSDs are celebrated for their ability to produce stable hysteretic behavior, owing to the inelastic deformation of their strips, alongside benefits such as lightness, ease of manufacture, and straightforward post-earthquake replacement. This research extensively examines SSD applications, design principles, and innovations in their modeling, optimization, and production processes. The literature highlights SSDs' consistent performance in resisting both compression and tension, their adaptability in strength, ductility, and energy dissipation through modifications in strip configurations and the superiority of non-prismatic and hourglass-shaped designs over traditional options. Numerical analyses have been conducted to assess the effectiveness of non-prismatic slit dampers in comparison to their prismatic counterparts within braced frames. Three distinct braced frame configurations have been analyzed: one with a diagonal brace without a damper, another featuring a uniform prismatic slit damper, and a third incorporating a non-prismatic slit damper with an hourglass shape. The analysis primarily compared these systems' hysteresis behavior, ductility, and energy dissipation capacities. Results indicate a significant enhancement in performance when utilizing non-prismatic slit dampers. Notably, these dampers exhibited a remarkable 69% increase in cumulative energy dissipation compared to prismatic ones. Furthermore, the study reveals that a steel slit damper-braced frame, when equipped with optimally designed slit geometries, can tolerate inter-story drifts in excess of 2% while simultaneously achieving a greater than 12% increase in energy dissipation efficiency.

Keywords: Steel Slit Damper; Braced Frames; Hysteresis Curves; Ductility; Energy Dissipation.

1. Introduction

The energy generated by large earthquakes must be mitigated using tools that enable structures to withstand intense ground motions. A significant portion of the transmitted energy can be absorbed, attenuated, and reflected by enhancing the damping capacity of the structure. As a result, the energy transfer to structural members decreases, reducing the energy consumption required for these elements [1–3]. Steel-yielding dampers are widely recognized as effective techniques for controlling structure vibrations [4–6]. Utilizing metallic dampers of this kind aims to concentrate inelastic deformations and prevent damage to other prominent frame members [7–9]. Among the steel-yielding dampers, steel slit dampers are the most commonly employed [10]. These dampers offer additional advantages such as lightweight construction, ease of fabrication, and convenient replacement following large earthquakes.

Structures may often suffer severe damage from wind, traffic, and earthquakes, among other dynamic and environmental loads [11]. Over 132 buildings were destroyed in the Mexico City earthquake, and over 200 were

* Corresponding author: zalsadoon@sharjah.ac.ae

doi) http://dx.doi.org/10.28991/CEJ-2024-010-04-019



© 2024 by the authors. Licensee C.E.J, Tehran, Iran. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC-BY) license (http://creativecommons.org/licenses/by/4.0/).

damaged in the Loma Prieta earthquake in Northern California in 1989 [12, 13]. There are several ways in which structural vibration controls can be used to reduce or avoid such damage, and these techniques have been successfully implemented in several different civil structures for this purpose [14–16]. Control systems that manage structural vibration are crucial in various engineering applications [17–19]. These systems are classified into four main groups based on their method of operation: passive, active, semi-active, and hybrid types [20–22]. Passive control systems are designed to dissipate or absorb vibrational energy without requiring external power input [23, 24]. Common passive control devices include tuned mass dampers [25]. Active control systems utilize sensors, actuators, and controllers to monitor and counteract structural vibrations in real-time [26]. Semi-active control systems combine elements of both passive and active systems [26]. Hybrid control systems integrate multiple control strategies, combining the advantages of different approaches to optimize performance and efficiency [27].

Fundamentally, the utilization of energy-dissipation mechanisms has been extensively studied and applied in diverse fields [28]. Among these mechanisms, viscous fluid dampers, friction dampers, viscoelastic dampers, and metallic yielding dampers effectively absorb and dissipate energy [29–31]. Viscous dampers force a fluid [32–35], while friction dampers utilize the frictional force between surfaces to dissipate energy. These dampers typically comprise sliding interfaces with high-friction materials or devices such as friction plates [36]. Metallic yielding dampers employ yielding elements, such as steel plates or rods, designed to undergo controlled yielding or plastic deformation under load [6, 37–39]. Finally, viscoelastic materials exhibit viscous and elastic behavior, making them well-suited for energy dissipation applications [40, 41].

Over the past four decades, it has become common practice to install passive dampers to improve structures' hysterical behavior and dissipate energy with no external power source required [42, 43]. By using these systems, the main elements of the building suffer less damage, and energy is dissipated more efficiently [44–46]. As well as maintaining stability over a wide range of temperatures, these dampers have extremely high fatigue resistance due to their low cycle and displacement rates, bolted attachments, low fabrication costs, tolerances for large displacements, and long-term durability [47–49]. These advantages of steel-yielding dampers have made them a popular passive energy dissipation device among all kinds of dampers. Several passive vibration control devices have been developed and implemented in buildings, including steel plates with slits or openings susceptible to shear [50–52].

Generally, slit dampers are considered metallic-yielding dampers because the yielding of their steel strips determines their damping behavior [53]. A new device, the multi-slit damper, combines weak and strong slit dampers in series to retrofit soft-first story structures against earthquakes [54]. Its energy dissipation occurs in two stages: weak-slit damper yielding during minor earthquakes and activation of the strong slit damper through a gap mechanism during major earthquakes. Experimental and analysis results confirm the device's effectiveness in controlling seismic response [54]. In a subsequent investigation, SSDs were employed, revealing that they substantially enhanced the lateral load capacity by up to 40% and ductility by 1.88 [55]. Additionally, experimental and numerical testing has indicated the potential of single-plate SSDs to be incorporated as a component of seesaw brace systems, as an addition to conventional bracing, or as a beam-to-column connection [56]. These SSDs have effectively reduced the forces and displacements experienced by structures during seismic events.

SSDs have been the main focus of numerous studies, and various methods and devices have been proposed based on necessities. Due to some special characteristics, architectural aspects, and requirements, the design and retrofitting of building structures with SSDs have been of great interest. Therefore, it is crucial to evaluate existing designs and configurations and analyze the benefits, limitations, enhancements, and potential optimizations that can be implemented in the future. The research thoroughly reviews commonly used slit dampers in civil structures to ensure greater adaptability and reliability, focusing on improving performance.

This study embarks on a pioneering endeavor to conduct a comprehensive review of steel slit dampers, an area which, to the best of the authors' knowledge, has not been exhaustively explored in existing literature. This gap underscores the imperative to delve into the evolution and applications of these dampers within structural systems. The review meticulously scrutinizes the influence of slit dampers on structural performance, integrating insights from contemporary research. A methodical appraisal and citation of scholarly articles from 1998 to 2023 is undertaken. This approach ensures an extensive coverage of all pertinent facets of slit dampers, encompassing their diverse applications, design considerations, configurations, integration within steel-braced frame structures, and potential avenues for enhancement. Moreover, this paper assessed numerically the comparative efficacy of non-prismatic versus prismatic slit dampers in structural frames. This analysis is conducted across three distinct structural frame models: a diagonally braced frame, a frame equipped with a uniform prismatic slit damper, and another featuring an innovatively designed hourglass-shaped non-prismatic slit damper. Building on the insights gleaned from recent technological advancements, the review culminates in proffering strategic recommendations for future research trajectories in this field.

The research paper is organized into seven sections. Section one discusses the prior art and the research gap in the literature, while section two outlines the methodology employed in this research. Section three presents an overview of steel slit dampers, including their concept and basic theoretical analysis. Following this, section four comprehensively

reviews various types of SSDs and discusses their applications and performance in braced frames. Section five conducts a comparative numerical analysis on three test frames using the finite element software ABAQUS. Specifically, one of these frames represents a special braced frame without a damper (SBF), while the other two frames are equipped with slit dampers—one utilizing a slit prismatic damper. Section six summarizes the conclusions drawn from the conducted research. Lastly, section seven outlines recommendations and proposes future avenues for applying SSDs in steel-braced frames, emphasizing their significance as crucial seismic fuses.

2. Research Methodology

In the field of building construction, slit dampers are recognized for their effectiveness in seismic mitigation through energy absorption and dissipation. These devices, composed of slender metallic strips, are integrated into the structure, playing a critical role during seismic events by mitigating the energy. This study reviews existing literature on slit dampers, focusing on aspects such as structural configurations, energy dissipation, hysteresis behavior, failure mechanisms, and ductility. The methodology, illustrated in Figure 1, includes a stringent review and selection process to identify relevant studies that shed light on the optimal design and implementation of slit dampers.



Figure 1. Flowchart depicting the methodology employed in this study

The aim is to assess the impact of slit dampers on the seismic resilience of buildings, leveraging data from prior research. Through a detailed analysis, this work synthesizes key findings and identifies best practices in the application of slit dampers for seismic protection. The outcome contributes to the body of knowledge by outlining strategies for enhancing building safety against seismic events.

3. Overview of Steel Slit Damper

3.1. The Concept of Slit Dampers

SSDs, or steel slit dampers, are widely recognized as one of the most prevalent passive dampers employed in structural engineering. These devices exhibit stable hysteretic behavior, highly resist temperature variations in the surrounding environment, demonstrate exceptional long-term reliability, and can be cheaply constructed [57]. Typically, SSDs are positioned between two structural elements or components, allowing for significant relative displacements. The energy generated during an earthquake can be effectively dissipated by subjecting to the cyclic movement of steel strips within the damper. Consequently, SSDs are extensively utilized in seismic design and earthquake retrofitting to ensure appropriate seismic response in structures. In the manufacturing process of SSDs, the damper plate can be perforated using various slits. Depending on the specific situation or element type, these slits can also be introduced in the web or flange of an I-section. In the manufacturing process of SSDs, the damper plate can be perforated using various slits. Depending on the specific situation or element type, these slits can also be introduced in the web or flange of an I-section. SSDs can be classified into two main categories based on their geometric characteristics: uniform prismatic and uniform non-prismatic slit dampers [58, 59]. Prismatic dampers maintain a consistent cross-sectional shape for all steel strips throughout the device's height, as their name implies. On the other hand, non-prismatic dampers exhibit varying cross-sectional shapes along their height (refer to Figure 2). A device in which all steel strips possess the same shape is considered uniform, whereas a device comprising different steel strip shapes is considered non-uniform, as depicted in Figure 3.



Figure 2. Traditional configurations of slit dampers and their lateral deformation; prismatic and non-prismatic [58]



Figure 3. An example of a non-uniform slit damper configuration [59]

Oh (1998) [60] introduced SSDs in the literature, as depicted in Figure 4. A similar concept was further investigated and implemented experimentally by González-Sanz et al. [58]. Their study focused on steel slit plate dampers subjected to shear deformation. They derived analytical equations that predict these dampers' ultimate energy absorption capacity. Additionally, they observed that the resulting energy could be analytically estimated, with their experimental findings closely aligning with the analytical predictions. The energy dissipated by these dampers arises from the flexural yielding of their strips during inelastic cyclic deformation [61-63]. Each damper strip undergoes double curvature bending upon sufficient story drift, forming plastic hinges at both ends. Numerous studies have also focused on developing this type of damper [64, 65].



Figure 3. The configuration of steel slit damper and deformation under the lateral load [60]

Finally, SSDs can provide extensive design flexibility, permitting adjustments to parameters such as strip count, configurations (uniform prismatic or non-prismatic and non-uniform prismatic or non-prismatic), and dimensions (height, width, and thickness). This adaptability facilitates utilization across diverse load scenarios, drift conditions, and energy dissipation requirements. The performance of the SSDs can be modified by adjusting these parameters. This configurational flexibility empowers designers with a spectrum of design options. The analytical equations in the next section reveal that these parameters exert notable influence on the performance of slit dampers, impacting factors such as strength capacities, elastic stiffness, and yield displacement.

3.2. Basic Theoretical Analysis

The governing equations for SSDs can be rigorously deduced from the principles of strength of materials. Figure 5 illustrates how the strip ends of the damper were idealized to predict the yield strength and deformation when the round-shaped ends were replaced with straight ones [53].



Figure 4. Idealization of strips

The flexural yielding loads at the ends of strips, denoted as $Q_{y,b}$, and the shear yielding loads of the strips, denoted as $Q_{y,s}$, represent the yield capacities of the slit plates. Additionally, the number of strips is represented by n. The total height of the strip is specified as H_t , the thickness of the slit gusset plate is denoted as t, the radius of the ends of the strips is indicated by r, and the strip width is denoted as B. The equivalent height, denoted as H' in Figure 5, is calculated using Equation 1.

$$H' = H + \frac{2r^2}{H_T}$$
(1)

Equation 2 is used to calculate the yielding displacement (δ_y) of the (SSDs) based on elasticity:

$$\delta_{y} = \delta_{b} + \delta_{s} = \frac{Q_{y} \cdot (\dot{H}^{3})}{nEtB^{3}} (1 + 3\ln\frac{H_{t}}{\dot{H}}) + \frac{3Q_{y} \cdot (\dot{H})}{2ntBG} (1 + \ln\frac{H_{t}}{\dot{H}})$$
(2)

Where δ_b and δ_s represent the displacement components due to flexural and shear deformations. Q_y refers to the yielding strength of the SSDs, which is determined as the minimum between the flexural yielding ($Q_{y,b}$) and the shear yielding loads of the strips ($Q_{y,s}$), as described in Equation 3, *E* denotes Young's modulus, and *G* represents the shear modulus.

It can be observed that the strips exhibit behavior consistent with that of a series of partially fixed-ended beams and undergo double curvature under small relative displacement between both support ends. Consequently, it is possible to determine the yielding and ultimate strength of the slit damper analytically by employing the following simplification [63]:

$$\begin{cases} Q_y = \min\{Q_{y,b}, Q_{y,s}\} \rightarrow \\ Q_y = \min\left\{n\frac{tB^2}{2H'}\sigma_y, n\frac{2tB}{3\sqrt{3}}\sigma_y\right\} \end{cases}$$
(3)

Where *n* is the number of strips, and σ_y is the yield stress of the slit plate.

Plastic hinges are formed at the ends of each strip when a significant displacement occurs. As a result, the calculation of the ultimate strength of SSDs is performed according to the following procedure [63]:

$$\begin{cases} Q_B = \min\{Q_{B,b}, Q_{B,s}\} \rightarrow \\ Q_B = \min\{n\frac{tB^2}{2H'}\sigma_B, n\frac{2tB}{3\sqrt{3}}\sigma_B\} \end{cases}$$
(4)

Where Q_B refers to the ultimate strength, which is determined as the minimum between the flexural strength ($Q_{B,b}$) and the shear strength of the strips ($Q_{B,s}$). σ_B is the ultimate stress of the slit plate.

Certain observations can be emphasized based on the research conducted by Chan & Albermani [61] on the seismic behavior of slit dampers. The ends of the strips were rounded, and the overall length of the strips was considered during their derivations. Consequently, the equations for determining the yield strength (P_y) and stiffness (K_d) can be expressed as follows:

$$P_y = n \frac{\sigma_y t b^2}{3l_0} \tag{5}$$

$$K_d = \operatorname{cn}\frac{Etb^3}{l_0{}^3} \tag{6}$$

The dimensions of the strips are denoted by the variables l_0 (length), t (thickness), and b (width). It is imperative to acknowledge that, to ascertain the elastic stiffness of the slit dampers (SSDs) via Equation 6, an underlying assumption is made regarding the strips' ends being partially fixed [61]. Moreover, the stiffness coefficient, c, was derived from

experimental data and utilized to calibrate the elastic stiffness. In this context, the radius of the fillet at the end of the strips can be considered another factor influencing the behavior of the SSDs. Previous research findings suggest that a minimum value for this parameter should be observed to mitigate the risk of premature fractures caused by stress concentration at the strip's end regions [66].

The analytical equations reveal that such as strip count and dimensions (height, width, and thickness) exert notable influence on the performance of slit dampers, impacting factors such as strength, elastic stiffness, and yield displacement. It becomes apparent from the analytical expressions that the structural attributes of the damper, encompassing yield strength (refer to Equations 3 and 5) and ultimate capacity (as depicted in Equation 4), exhibit a linear correlation with the thickness of the slit damper. Consequently, as the thickness of the slotted plate increases, both the yield strength and ultimate strength will also increase linearly. In contrast, according to Equations (3 to 5), an inverse correlation exists between the length of steel strips and strength capacities. As the length of the strip increases, the yield and ultimate strength of the damper decrease. Additionally, widening the slit damper, as indicated by Equation 3 and 4, results in an augmentation of both yield and ultimate strength.

On the other hand, SSDs are known to be thin slit plates. Therefore, to ensure SSDs exhibit stable hysteretic behavior and effectively resist both compression and tension, the slit plates, particularly the strips within these dampers, must avoid out-of-plane buckling. Therefore, it is crucial to carefully select the dimensions of the strips, including their length, width, and thickness. Specifically, the slenderness ratios of length to width and width to thickness must adhere to the criteria to prevent buckling under critical conditions.

3.3. Summary

The strips' thickness, height, and width are crucial factors that significantly influence the strength capacities, stiffness, and other key parameters of slit dampers. It should be emphasized that appropriately modifying these parameters can improve the seismic performance of slit dampers. Moreover, when the bending behavior governs, slit dampers generally exhibit stable performance levels that are deemed acceptable. Finally, these equations afford a computational avenue to discern pivotal mechanical parameters intrinsic to the SSD system, thereby preventing the prerequisite for exhaustive simulation and empirical testing.

4. Application of the Steel Slit Damper in Braced Frames

Numerous earthquakes have recently occurred worldwide, significantly damaging buildings and bridges. To withstand lateral loads, concentrically and eccentrically braced frames have been widely employed [67-69]. These frames have desirable stiffness and strength, making them suitable for earthquake-resistant structures [70, 71]. However, it should be noted that CBFs exhibit lower ductility compared to moment frames and eccentrically braced frames. This system dissipates energy through the inelastic deformation of braces, which can buckle, yield, and undergo inelastic deformation following a major earthquake [72-74]. Extensive research has been conducted to enhance the seismic performance of CBFs. For instance, investigations have focused on preventing brace buckling through restraining mechanisms, examining connection details of corner and middle gusset plates, evaluating the type and size of braces, and optimizing the arrangement of the braces [75-77]. For instance, a buckling-restrained brace (BRB) is engineered and employed to bolster strength, stiffness, and ductility against lateral loads. BRBs mitigate structural damage and augment the braced structures' seismic resilience [78, 79].

Recently, several studies have been conducted to address enhancing brace performance by incorporating energy dissipation devices. This approach aims to absorb the plastic action and prevent damage to other crucial components [80-84]. Significant advancements have been made in the system based on the "weak gusset plate-strong brace member" concept. Various techniques have been proposed to implement this concept, such as utilizing gusset plates with a low yield point. For instance, Chen & Chang [39] investigated the impact of gusset plates with low yield points. Their findings demonstrated a reduction in the pinching effect of braced frames resulting from corner gusset plate buckling.

Benavent-Climent [84] conducted experimental studies on slit plates subjected to shearing deformations to assess their ultimate energy absorption capacity. The study considered three parameters: strip width to plate height (B/H) ratio, material properties, and the loading pattern. Based on the assumptions made, the ultimate energy absorption capacity was determined by combining the energy dissipated by the structural skeleton and the Bauschinger component. In 2002, Lee et al. [83] proposed and developed an alternative device for efficient and cost-effective dissipation of seismically-induced energy. This device, known as a hysteric plate slit damper, could be easily installed in cross-braced frames and was primarily subjected to shear forces (see Figure 6). Subsequently, it became the first steel slit damper utilized in a V-braced system [60]. Another energy dissipation device, known as a tube-in-tube damper (TTD), was introduced by Benavent-Climent for braces [84]. TTDs involve cutting slits through the device's outer hollow section, generating strips attached to the inner section through welding. The strips dissipate energy through flexural/shear yielding when TTDs move toward the brace. A schematic view and detailed illustration of the TTD are presented in Figure 7.



Figure 5. Slit plate damper in a cross-braced frame [83]



Figure 7. TTD as brace-type seismic damper [84]

To address the strain concentration at the ends of strips and determine the optimized shape of the slits, Ghabraie et al. [85], by utilizing the bidirectional evolutionary structural optimization technique, were able to optimize the shape of the slits and develop an effective configuration. The device's energy dissipation capacity and fatigue resistance under low cycling conditions were significantly enhanced by incorporating this configuration. In addition, A numerical model was used by Aminzadeh et al. [86] to obtain optimal boundary shapes for slit dampers. They employed the iso-geometric analysis method, allowing precise modeling of complex geometries for nonlinear damper analysis. Using B-splines with non-uniform rationality, an optimized shape was generated to simulate a conventional steel slit damper under a volume constraint. The newly proposed steel slit damper was compared with other shapes from the literature, and an optimization process was proposed and evaluated based on its performance. The optimal shape demonstrated a 12% increase in energy dissipation. The stress distribution in the proposed damper and the compared dampers can be seen in Figure 8.

González-Sanz et al. [58] investigated the development of a mild steel damper with non-uniform vertical slits. The study aimed to analyze the relationship between different forms of vertical slits on the core energy plate and their ability to dissipate energy and resist buckling. The researchers formulated theoretical equations to determine key damper parameters and validated numerical models using experimental data. By analyzing earthquake time-history data, they demonstrated the effectiveness of the dampers in reducing seismic responses of the prototype frame and achieving satisfactory energy dissipation. In another study, Tagawa et al. [87] utilized SSDs in a seesaw system, where the bracing members remain in tension during lateral movements (see Figure 9). The researchers conducted cyclic loading tests on the framed structure with this proposed system and found that it exhibited stable hysteretic behavior and had substantial energy dissipation capacity. Furthermore, all specimens showed yielding of the slit dampers at a story rotation angle of 0.001 rad.



Figure 6. Stress distribution in slit dampers with various configurations; (a) conventional slit dampers [86], (b) optimal shape [86], and (c) proposed shape by Ghabraie et al. [85]



Figure 7. General and deformed configurations of a seesaw system equipped with SSDs [87]

Furthermore, Tagawa et al. [87] implemented slits as dampers in chevron-braced frames. A separate investigation by Ahmadie Amiri et al. [47] found that incorporating a block slit damper (BSD) can enhance chevron frames' performance. The BSD consists of a steel block with multiple slits, and it offers several desirable characteristics, such as high shear strength, energy absorption and dissipation capabilities, and favorable economic implications. The BSD can be integrated into a chevron frame, as Figure 10(a) illustrates. Building upon the previous study by Katal Mohseni et al. [88], further optimizations were made to steel BSDs to enhance their energy absorption capacity. Various geometric dimensions were examined, and the findings indicated that the frame equipped with BSDs exhibited significantly improved ductility and stiffness [88] compared to the earlier study [47]. Consequently, the energy absorption during hysteresis cycles experienced a substantial increase.



Figure 10. Braced frames equipped with BSD and SS-TTD dampers; (a) BSD device configuration [47] and (b) View of the SS-TTD in general and a typical building frame [58]

In a study by Kim [89], an investigation was carried out on a box-type SSD and a multi-SSD system. Gonzalez-Sanz et al. [58] explored a novel type of stainless-steel tube-in-tube damper (SS-TTD) composed of two telescopic tubes. The energy dissipation in the brace-type damper occurs through flexural plastic deformations of the strips when subjected to forced axial displacements. The exterior tube walls of the SS-TTD feature a series of slits that form the strips, which are connected to the interior tube (refer to Figure 10(b)). The seismic behavior of the SS-TTD was evaluated using quasi-static and dynamic shaking table tests. Comparative analysis revealed that the SS-TTD exhibits approximately four times higher cumulative ductility than slit-type plate dampers. Furthermore, its dissipation capacity is three times that of mild steel dampers and sixteen times that of high-strength steel dampers. Guo et al. [90] proposed an adjustable steel damper, along with potential installation locations in frames. To achieve maximum plasticity and fully use the load-bearing capacity of the steel strips, they must yield simultaneously and distribute energy uniformly across the entire section. The pumping hysteresis loops observed in the strips effectively dissipate potential energy. The bending moment strength primarily depends on the width of the central region.



Figure 8. Illustration of the proposed damper and the location of potential installations in frames [90]

As an alternative to conventional link beams, Askariani et al. [66] introduced the concept of a slit link beam to enhance the structural performance of eccentrically braced frames (EBFs). Vertical strips can be formed by incorporating perforations in the web to create the link beam (refer to Figure 12(a)). The authors derived formulas to analyze the characteristics of the slit link beam and validated their findings using the ABAQUS software. Askariani also proposed a concept involving replaceable components [91]. Numerical simulations were conducted on a structure equipped with a novel brace-type slit damper (BSD), evaluating the influence of geometrical parameters on the frame's performance. The slit plate has two diagonals on each side, with links designed and arranged based on a specific geometry (see Figure 12(b)). A gusset plate is utilized to restrict the out-of-plane displacement of the diagonal members at one end, while a pin is employed at the other end. It was observed that all examined frames exhibited stable hysteresis curves under various loading conditions. Furthermore, upon completion of loading, no cracks were detected in the damper links. The results indicate that the proposed damper is an effective energy-dissipating device suitable for implementation in building structures. In subsequent research, Zhao et al. [92] developed a new type of damper known as the perforated web H-type brace damper (PWHBD). The H-profile brace used in their damper system features multiple slits along its length (see Figure 13), allowing for seismic energy dissipation through flexural yielding of the perforated webs under in-plane shear. The hysteretic curves exhibited stable and substantial behavior, further supporting the effectiveness of the damper in dissipating energy during seismic events.



Figure 12. Eccentrically and concentrically braced frame equipped with slit dampers; (a) eccentrically braced frame with slit link beam [66] and (b) brace-type slit damper [91]



(a)



(b)

Figure 13. An example of a PWHBD test specimen [92]; (a) specimen dimensions and (b) view of the overall test setup

Based on a study by Javidan et al. [93], a steel multi-slit damper (MSD) has been proposed as a retrofit solution for partition walls. The study involved experiments on reinforced concrete frames comprising two stories, both with and without the suggested dampers. The study's findings indicate that implementing the MSD retrofit strategy enhances the hysteretic behavior of the structures. Similarly, Benavent-Climent et al. [94] have developed a novel metallic damper employing mild steel plastic deformation (refer to Figure 14). This damper is designed to dissipate energy during severe or extreme earthquakes. Including a gap mechanism in the damper helps prevent high-cycle fatigue damage. Additionally, the damper exhibits a remarkable energy dissipation capacity when subjected to large deformations caused by extreme ground motion. Dynamic shake-table tests were conducted on dampers installed in reinforced concrete structures alongside static cyclic tests of dampers isolated from the structure to evaluate their performance. The test results have enabled the development of a hysteresis model that accurately predicts the force-displacement curve of the damper under different cyclic loadings. The proposed hysteresis model can effectively capture increased stiffness and strength at large deformations. Furthermore, dampers' energy dissipation capacity is influenced by the phase in which they fail, suggesting new equations for predicting this behavior. The test results and a simple numerical model demonstrate that the damper exhibits a stable hysteresis response, predictable cyclic behavior, and a reliable ultimate energy dissipation capacity.



Figure 14. Details of the proposed damper by Benavent-Climent et al. [94]

According to a recent study by Almohammad-albakkar et al. [95-97], a novel damper that enhances the seismic performance of X-CBFs (Cross-Braced Frames) has been developed and tested numerically and experimentally. The proposed design consists of two cross braces with two segments each, where a grooved gusset plate damper (GGPD) was utilized to connect the four segments at the bay's midpoint. This connection prevents plastic action or buckling in the braces and other structural members. Figure 15 illustrates the addition of stiffener plates to the inner and outer edges of the GGPD, which effectively limits the occurrence of local instability in the grooved gusset plate. Specifically, it prevents early-stage out-of-plane buckling during story drift.



Figure 15. The grooved gusset plate damper (GGPD) [95]; (a) installation of GGPD in a cross-braced frame, (b) configuration of GGPD, and (c) the distribution of plastic strain

The working mechanism of the GGPD relies on the yielding of the gusset plate strips in in-plane bending, resulting in the dissipation of seismic energy. The experimental results indicate that the innovative system (GGPD) can tolerate a more than 3% relative drift. Additionally, a computational simulation was performed using the finite element software ABAQUS to assess the performance of this novel system. Additionally, the yield and ultimate capacities of the damper were analytically calculated using derived formulas. A comparison was made between the numerical outcomes and the results obtained from the analytical equations. The comparison results demonstrate a good agreement between the numerical and analytical findings.

In a recent study by Heyrani & Shooshtari [98], the hysteresis performance of new SSDs under cyclic loading. FEA was employed to simulate different slit sizes and the number of slits in single-column and dual-column SSDs. Sample SSDs with the highest ultimate strength were then subjected to testing. The findings indicate that dampers featuring steel strips of elliptical openings demonstrate enhanced energy dissipation compared to the tested specimens. Dual-column SSDs also show higher initial strength than single-column SSDs. In addition, single-column SSDs exhibit stable hysteresis performance, minimizing strength loss during displacement cycles. Rousta et al. [3] introduced vertical steel panel flexural yielding dampers (VSPFYDs) to enhance the cyclic behavior and overall performance of SSDs (Figure 16). Numerical methods using ABAQUS software and extensive parametric studies were employed to validate and investigate their study. Comparison with traditional SSDs confirmed that VSPFYDs exhibited improved seismic parameters. Notably, VSPFYDs with a height of 15 cm demonstrated superior performance in various aspects. The plate connection construction was found to have a significant impact on the performance and hysteresis behavior of VSPFYDs compared to SSDs.



Figure 16. Schematic view and geometric parameters of the proposed system by Rousta et al. [3]; (a) VSPFYD and (b) SSD

Hui et al. [99] introduced and conducted experimental assessments on an economical double-sided slotted steel tube shear damper (refer to Figure 17). This device leverages elastic-plastic deformation within the plane of the slotted steel plate on the steel tube's side to absorb seismic energy, effectively mitigating vibrations. Quasi-static tests were systematically conducted to investigate the influence of various design parameters on operational efficiency, energy dissipation capacity, and failure characteristics. The test results demonstrated the damper's robust plastic deformation capabilities, outstanding seismic performance, and substantial energy dissipation capacity. The hysteretic curve exhibited both symmetry and completeness, resembling a shuttle shape. Notably, increasing the bending element's width and the steel tube's wall thickness yielded significant improvements in energy dissipation.



Figure 17. Double-sided slotted steel tube shear damper [99]

In the interest of comprehensiveness, Kang et al. [100] developed a seismic retrofit scheme using a combination of a steel moment frame and SSDs to mitigate earthquake damage in buildings. The hybrid system underwent a cyclic load test within an existing reinforced concrete frame. The device demonstrated energy dissipation capabilities similar to slit dampers, with a slight reduction in stiffness compared to traditional steel frame reinforcement methods. The results of a finite element analysis validated the test outcomes, showing a significant enhancement in structural strength and ductility when using slit dampers compared to bare structures. The model structure effectively reduced its seismic response below the target limit state by implementing slit dampers. The stress distribution in the test structures, obtained from the finite element analysis, is depicted in Figure 18.



Figure 18. The stress distribution in the test structures resulting from the finite element analysis [100]

Traditionally used slit dampers are prone to cracking and breaking because of stress buildup at the ends of the strips when used for earthquake retrofitting. Bae et al. [101] proposed an improved damper design using radius-cut cokeshaped strips to address this issue and enhance ductility. Figure 19 illustrates the von Mises stress distributions for the three analysis models. By applying the reduced beam section method, the height-to-width ratio of the strip was increased to induce more bending deformation than shear deformation. Plastic hinges were intentionally created by focusing stress on the radius-cut section. This design specimen has higher inelastic deformation capacity and reduced fracture fragility. The experimental approach resulted in a finite element analysis demonstrating the coke-shaped damper's improved ductility. During cyclic loading, the radius-cut section was identified as the location of the final fracture, indicating sustained energy dissipation. The hysteresis curves of this damper are presented in Figure 20.



Figure 19. Radius-cut coke-shaped strip damper [101]; (a) Reduced beam section models (plastic hinges number: 0, 2, and 4ea/strip) and (d) An explanation of the strip with four plastic hinges



Figure 20. Hysteresis curves of radius-cut coke-shaped strip damper [101]; (a) first specimens (without reduced beam section) and (b) second specimen (with two plastic hinges)

A new double C-section steel slit damper (DCSSD) was introduced to examine the impact of strip aspect ratio, flange thickness, damper length, and steel grade on its resistance and hysteretic behavior [102]. Experimental results revealed that the DCSSD exhibited favorable structural characteristics. All DCSSD specimens displayed an equivalent damping ratio above 0.45, with Q160 exceeding this threshold and achieving a damping ratio exceeding 0.50. Moreover, the DCSSD utilizing Q160 demonstrated a cumulative displacement of approximately 1500 mm. Numerical models were subsequently developed for the test specimens, demonstrating excellent agreement with the experimental findings. He et al. [103] studied the structural performance of steel slit shear panels (SPs) in MRSF structures. The time-history analysis results indicated that installing SPs reduced floor displacement responses but increased floor acceleration responses. Among different SP types, the low-yield SP was the most effective in reducing maximum and residual interstory drift.

In recent years, Block Slit Dampers (BSDs) have emerged as a promising solution for passive structural control, particularly in high seismicity regions where highly ductile systems like SMRFs are required [104]. BSD devices with secondary hardening in their hysteretic curves have demonstrated the ability to meet performance objectives for flexible steel SMRFs in such regions. Additionally, BSD devices effectively reduce inter-story drift ratios and regulate absolute floor accelerations (AFAs), minimizing damage to non-structural elements. Thus, BSD devices offer a suitable solution for designing and retrofitting flexible steel SMRFs, providing hysteretic behavior and enhanced performance. Two experimental boundary conditions were employed in another study by Oh and Park [105] to examine the performance of this particular damper: i) shear loading alone and ii) combined shear and tensile loading. The combined shear-tensile loading resulted in a butterfly-shaped hysteresis curve for the damper, while shear-only loading produced a parallelogram-shaped curve [105]. The damper's properties are influenced proportionally by adding additional tension, contingent upon the damper's aspect ratio and maximum deformation angle. According to the cyclic hysteresis analysis, the damper's plastic deformation capacity remains similar under both tensile and exclusive shear loading conditions. Furthermore, the energy absorbed up to failure is equal to that observed when the damper is only subjected to shear loading. However, when subjected to additional tensile loading, the damper dissipates a notable amount of unstable energy due to out-of-plane deformation. The damper exhibits lower hysteresis under additional tension than the predicted curve used for evaluating plastic energy based on test results.

Non-buckling slit dampers (NBSDs) integrated into a window-type seismic control system (WSCS) offer a viable solution for developing an effective seismic control system for existing reinforced concrete buildings [106]. Buildings retrofitted with NBSD-WSCS frames demonstrated minimal damage during seismic events, even under the highest recorded seismic intensity in Korea, reaching approximately 300 cm/s², resulting in only minor or moderate damage. Lee et al. [106] successfully developed and validated NBSD-WSCS as an efficient method for seismic retrofitting.

4.1. Summary

According to existing research, the incorporation of steel slit dampers in braced frames has been found to enhance the overall ductility of structures. In particular, using SSDs has demonstrated the ability to achieve an inter-story drift ratio of over 2% when implemented in braced frames. The failure mechanism associated with these dampers typically involves the failure of the strips at their ends. Consequently, optimizing the configuration of these damper strips can enhance their performance and delay fracture. Significantly, the utilization of an optimized configuration not only increases the energy dissipation capacity but also improves fatigue resistance. Optimal-shaped SSDs have the potential to achieve an energy dissipation improvement of over 12%. Comparatively, non-prismatic and hourglass configurations have exhibited greater efficiency than conventional configurations. In terms of the hysteresis curve configuration, it is generally observed that this type of damper shows a parallelogram-shaped curve when subjected to shear loading.

5. Numerical Analyses and Discussion

5.1. Description of Frames and Dampers

The present study is dedicated to conducting a comparative analysis of three test frames. Specifically, one of these frames is a special braced frame without a damper (SBF), while the remaining two frames are equipped with slit dampers.

One of the frames is of slit prismatic damper (FSPD), while the other features a non-prismatic damper (FSNPD). In this study, the damper is affixed to the lower end of the brace, ensuring consistency across all three samples by maintaining meticulous dimensions and sections for the primary structural element within the braced frame. These dimensions include a height of 3.0 meters and a span of 3.5 meters. Subsequently, a comprehensive comparison is carried out, encompassing the evaluation of hysteresis curves, ductility, and dissipated energy among the results of the three numerical analyses. For visual reference, Figure 21 displays the frame used in this research. Additionally, Figure 22(a) and Figure 22(b) provide detailed illustrations of the uniform prismatic and non-prismatic slit dampers utilized in the other frames, respectively. The steel frame elements consist of an HSS steel section of $100 \times 100 \times 6$ for the brace, an IPE220 for the beam, and an IPB140 for the column.



Figure 9. Details of the braced frame, along with the proposed damper





5.2. Modeling Techniques

The numerical investigations in this study, which encompassed the analysis of all specimens, accounting for both geometric and material non-linearities, were conducted using the nonlinear finite element (FE) software ABAQUS

[102]. A 4-node doubly curved thick shell element with reduced integration (S4R) was chosen to represent the test specimens accurately. This approach effectively captures the intricate structural behavior under examination.

A sophisticated combined cyclic model, which incorporates both kinematic and isotropic hardening components, was employed to simulate the cyclic response of the material, guided by the Von Mises yield criterion. It is worth noting that this material model has previously undergone validation for its efficacy in replicating the inelastic behavior of steel, as evidenced in references [55, 56]. The FE models were constructed using St37 steel, and the elastic-plastic response of this material was modeled through a comprehensive nonlinear approach that integrates isotropic and kinematic hardening properties. The mechanical properties of the material are systematically outlined in Table 1.

Table 1. Mechanical characteristics for FEA models

Material	F _y (MPa)	F _u (MPa)	E (GPa)	v
St37	240	370	200	0.3

It is essential to emphasize that the FE modeling framework in this study intentionally excluded considerations of fracture evolution and material degradation. The bases of the specimens were modeled as pinned connections to effectively constrain any degrees of freedom that might result in unintended movements. Out-of-plane deformation was prevented at the upper column ends to mitigate out-of-plane buckling of the frame induced by lateral loads. The mesh size for the dampers was set at 5 mm, while other frame elements were adjusted to approximately 20 mm, as illustrated in Figure 23. The testing protocol for specimen examination adhered to the SAC guidelines [103], as shown in Figure 24. The testing regime was rigorously conducted up to a drift level of 3%.



Figure 11. Mesh configuration of the test frames: (a) SBF and (b) FSPD and FSNPD



Figure 12. Protocol for cyclic loading

5.3. Results and Discussion

5.3.1. Force-Displacement Relationship

The force-displacement curves derived from finite element (FE) models are illustrated in Figure 25 for three configurations: a frame without steel slit dampers (SSD) and frames equipped with SSD dampers, one encompassing prismatic and the other non-prismatic slit dampers. The analysis indicates that the maximum positive and negative load capacities of the frames with SSDs are closely aligned with those without SSDs. However, a noteworthy distinction emerges in their hysteresis behavior. The frame with no SSD exhibited diminished strength under compressive loads. This reduction is primarily attributed to brace buckling at a peak displacement of 6.7 mm, corresponding to a relative drift of 0.22%. Conversely, frames integrated with SSD devices showcased enhanced hysteresis behavior without any signs of instability. Notably, these frames achieved up to a 3% drift (90 mm) without a concomitant decrease in strength, a significant improvement over the non-damper frames. This finding is consistent with the outcomes reported by Almohammad-albakkar et al. [95, 96], who observed that the cross-braced frames equipped with a steel slit damper achieved a relative drift exceeding 3%. A distinctive butterfly shape, indicative of improved energy dissipation capabilities, characterizes the hysteresis curves for damper-equipped frames. The butterfly-shaped hysteresis curves observed in the braced frame with SSDs closely resemble those documented in the literature for block slit dampers [47]. Further analysis reveals a marked difference between the frames equipped with different types of dampers. Frames with non-prismatic slit dampers exhibited a more pronounced hysteresis curve, especially in the initial cyclic loading phase, while maintaining comparable strength to those with prismatic slit dampers.



Figure 13. Force-displacement relationship

Table 2 presents a detailed comparative summary of these findings. It shows that the ultimate strength of the frame without a damper (SBF) marginally exceeds that of the frames with SSDs. In the positive phase, at a maximum displacement of 90 mm, the SBF attained a maximum load of 338.3 kN, compared to 307.6 kN and 314.7 kN for frames with prismatic and non-prismatic slit dampers, respectively. On the contrary, during the negative phase, the SBF achieved its highest load of 272.8 kN at a displacement of 6.72 mm, whereas frames FSPD and FSNPD exhibited peak loads around 299.1 kN and 307.2 kN at a displacement of 90 mm, respectively. The table also shows the comparable test results of frames' elastic stiffness and energy dissipation, discussed in the following sections. Furthermore, when comparing the outcomes derived from FSNPD with those of FSPD, it is evident that the non-prismatic shape boosts elastic stiffness by more than 25% compared to the prismatic configuration.

Table 2. Comparison of the ultimate stren	igth.	ductility	v. and	energy	dissi	pation	of the	test	frames
	O 2			· · •					

Test from s	Ultimate strength (kN)		Electic stiffness (IN/mm)	Energy dissipation		
1 est frames	⁺ Pu	'Pu	Elasuc sunness (kiv/iiiii)	(kJ)	Ratio with (SBF)	
Special braced frame (SBF)	338.27	-272.81	41.39	141.99	1	
Frame with slit-prismatic damper (FSPD)	307.62	-299.07	26.55	178.46	1.26	
Frame with slit-non-prismatic damper (FSNPD)	314.71	-307.22	33.73	240.17	1.69	

In terms of hysteresis behavior, both prismatic and non-prismatic slit dampers exhibit a butterfly-shaped curve, albeit with noticeable differences. The hysteresis curves of non-prismatic slit dampers appear broader, particularly during the initial loading stages. Consequently, non-prismatic slit dampers dissipate energy more effectively compared to prismatic dampers, with the former exhibiting approximately 35% greater dissipation. Regarding ductility, both damper types boast a similar ductility level, around 3%, as per the obtained results. However, it should be noted that according to prior experimental research [85], the use of a non-prismatic slit damper can notably enhance the damper's performance under fatigue.

5.3.2. Distribution of Plastic Strain

Figures 26 and 27 show the distribution of Von Mises stress across the evaluated frames. A noticeable reduction in stress is observed in the frame outfitted with steel slit dampers (SSD) devices, particularly evident at panel zones and gusset plate junctions. In contrast, the frame without SSD devices exhibited stress intensities approaching the material yield threshold, predominantly at the panel zones and the base of columns. This disparity underscores the efficacy of the SSD intervention in mitigating stress concentrations in frame structures subjected to cyclical or dynamic loadings. Additionally, Figures 27 to 29 show that both FSPD and FSNPD exhibit significant strain and stress concentrations at the root of the strips. Specifically, the FSNPD strips exhibit a more uniform stress distribution at their root, contrasting with the variations observed in FSPD. This observation underscores the influence of the non-prismatic shape of the strips on their ability to enhance the damper's performance, particularly in absorbing and dissipating energy.



Figure 14. The Von Mises stress distribution for the frame without damper (SBF)



Figure 15. The Von Mises stress distribution for the frame equipped with non-prismatic slit damper (FSNPD)

Further explanation is provided in Figures 28 and 29, which compare the deformation shapes of frames equipped with and without SSD devices under lateral load conditions. The plastic strain is more uniformly distributed across the dampers in the frame incorporating SSD. This phenomenon can be attributed to the inherent design characteristics of the SSD devices, which ensures that the steel strips within the dampers yield preferentially. Consequently, this preemptive yielding predicts the emergence of pronounced stress concentrations at other critical junctures within the frame structure.



Figure 16. Distribution of equivalent plastic strain (PEEQ) within the slit damper of the prismatic damper (FSPD)



Figure 17. Distribution of equivalent plastic strain (PEEQ) within the slit damper of the non-prismatic damper (FSNPD)

5.3.3. Energy Dissipation Analysis

Figure 30 presents a detailed depiction of the cumulative energy dissipation in the frames. This figure distinctly highlights the enhanced energy dissipation capacity of frames when integrated with a non-prismatic slit damper. Specimen FSNPD demonstrates greater stability and superior energy dissipation capabilities than the other specimens. Complementing this graphical representation, Table 2 provides quantitative insights into the cyclic energy dissipation observed in these frames.



Figure 18. The cumulative energy dissipation of the frames

A notable observation from this data is the disparity in energy dissipation between the braced frame without the slit damper and those outfitted with prismatic and non-prismatic slit dampers. The quantitative analysis reveals an increase in energy dissipation by 26% and 69% for frames with prismatic and non-prismatic slit dampers, respectively. Such findings indicate the substantial impact that damper design can have on the energy dissipation characteristics of frames. Additionally, upon comparing the results obtained from FSNPD with those from FSPD, it becomes evident that the non-prismatic shape enhances energy dissipation by over 35% compared to the prismatic shape. These findings are consistent with those of previous studies on slit dampers in the literature. The previous research revealed that utilizing a tapered shape can enhance energy dissipation by approximately 37% compared to the conventional initial shape in literature [85] and around 12~50% in the literature [86].

5.4. Summary

The results derived from the finite element analysis provide compelling evidence for the efficiency of Steel Slit Damper (SSD) devices in enhancing the cyclic performance of frame structures under lateral loading conditions. These devices, notable for their simplicity, low mass, and cost-effectiveness, demonstrate a significant capacity to attenuate lateral loads. This is evidenced by the force-displacement hysteretic curves generated for the frame equipped with SSD devices, which markedly outperform the bare frame. Notably, the frame incorporating non-prismatic slit dampers exhibited a superior cumulative energy dissipation compared to its counterparts. Quantitatively, the integration of non-prismatic dampers resulted in a 69% increase in the cumulative energy dissipation of the frame, highlighting the potential of these devices in structural engineering applications.

6. Conclusions

This study comprehensively evaluates Steel Slit Dampers (SSDs) in steel-braced frames, focusing on their consistent hysteretic behavior, customizable configurations, and significant enhancement of seismic resilience. Comparative analysis has been conducted on a Special Braced Frame (SBF) without dampers and two other frames, one with prismatic and the other with non-prismatic SSDs dampers. The following can be concluded from the research conducted:

- Slit dampers used in braced steel frames provide significant strength, ductility, and dissipation capabilities, and they provide stable and predictable hysteretic behavior, displaying uniform force resistance in tension and compression.
- SSDs performance can be tailored by their geometry by adjusting the strips' number, dimensions, and configuration. Notably, SSDs with non-prismatic and hourglass shapes exhibit superior characteristics. Damper designs featuring steel strips as elliptical slits demonstrate the highest energy absorption.
- Optimal-shaped non-prismatic SSDs have the potential to achieve an energy dissipation improvement of over 12% compared with conventional non-prismatic dampers. Furthermore, optimized dampers exhibit significantly larger cumulative displacement than conventional SSDs, indicating a higher inelastic deformation capacity.
- The installation of SSDs in braced frames has demonstrated their ability to withstand an inter-story drift ratio of up to 2%. Through experiments and numerical analyses, a non-prismatic damper has been identified as exhibiting greater ductility.
- When subjected to combined shear and tensile loading, SSDs of prismatic and non-prismatic dampers exhibit a butterfly-shaped hysteresis curve, whereas shear-only loading results in a parallelogram-shaped curve.
- In the conducted numerical analysis, it was recognized that the frame outfitted with non-prismatic slit dampers exhibited an improved cumulative energy dissipation, surpassing that of the frame with prismatic configuration. Quantitatively, this enhancement in energy dissipation was calculated to be 69%, signifying a substantial increase attributable to the incorporation of the non-prismatic damper.
- The numerical results further demonstrate an enhancement in the strength during the negative phase of loadings for frames equipped with FSPD and FSNPD.

The findings of this study conclusively demonstrate that the integration of dampers into structural designs significantly improves their capability to withstand high levels of relative drift, thereby enhancing their ductility. This increased ductility significantly reduces the seismic forces exerted on the structure, promoting cost-effective design strategies. Buildings enhanced with damping systems can maintain critical performance standards necessary for immediate occupancy (IO) and life safety (LS) in the aftermath of seismic events.

A key insight from this research is the localization of damage to the dampers, which significantly reduces the economic impact associated with post-earthquake restoration efforts. Consequently, the strategic implementation of damping mechanisms in building design is highlighted as an effective measure for enhancing structural durability and safety in earthquake-prone areas while offering a cost-effective solution for initial construction and subsequent earthquake recovery processes.

6.1. Recommendation and Future Work

The research provides a comprehensive review and numerical analyses of the use of steel slit dampers (SSDs) in braced frames, identifying several critical avenues for future research. These recommendations are pivotal for advancing our understanding of SSDs in seismic-resistant design. However, to enhance the academic rigor and clarity of the recommendations, certain modifications and expansions are suggested:

- It is highly commendable to explore the advantages of using alternative SSD materials for SSDs, such as low mild yield and stainless steel.
- Investigate SSD behavior under low-cycle fatigue, as SSD failures can be attributed to concentrated stress at the strip ends.
- Investigating the use of hybrid systems that combine slit dampers with other types of dampers is an insightful approach.
- Conducting comprehensive research on the behavior of SSDs under various force combinations, including shear force with moment and/or axial force, is essential. Specifying force combinations or scenarios of significant interest, including extreme loading conditions or varying frequency and amplitude, would be advantageous.

The recommendations above would offer a more targeted direction for future investigations.

7. Declarations

7.1. Author Contributions

Conceptualization, Z.A. and M.A.; methodology, Z.A. and M.A.; software, M.A.; validation, Z.A. and M.A.; formal analysis, Z.A. and M.A.; investigation, Z.A. and M.A.; resources, Z.A. and M.A.; data curation, M.A.; writing—original draft preparation, Z.A. and M.A.; writing—review and editing, Z.A. and M.A.; visualization, Z.A. and M.A.; supervision, Z.A.; project administration, Z.A. and M.A.; funding acquisition, Z.A. All authors have read and agreed to the published version of the manuscript.

7.2. Data Availability Statement

The data presented in this study are available on request from the corresponding author.

7.3. Funding

The authors gratefully acknowledge the financial support from the Sustainable Construction Materials & Structural Systems Research Group of the Research Institute for Sciences and Engineering (RISE) at the University of Sharjah.

7.4. Acknowledgements

The authors express their gratitude to the University of Sharjah and the Isfahan University of Technology for granting access to their facilities and for the support provided throughout writing this manuscript.

7.5. Conflicts of Interest

The authors declare no conflict of interest.

8. References

- Meshkat Rouhani, V. R., Zamani Ahar, G., & Saeed Monir, H. (2022). Experimental and numerical study on a new double-walled tuned liquid damper. International Journal of Engineering, 35(1), 29–44. doi:10.5829/IJE.2022.35.01A.04.
- [2] Jaisee, S., Yue, F., & Ooi, Y. H. (2021). A state-of-the-art review on passive friction dampers and their applications. Engineering Structures, 235, 112022. doi:10.1016/j.engstruct.2021.112022.
- [3] Rousta, A. M., Gorji Azandariani, M., Safaei Ardakani, M. A., & Shoja, S. (2022). Cyclic behavior of an energy dissipation system with the vertical steel panel flexural-yielding dampers. Structures, 45, 629–644. doi:10.1016/j.istruc.2022.09.047.
- [4] Javidan, M.M., & Kim, J. (2020). Steel hysteretic column dampers for seismic retrofit of soft-first-story structures. Steel and Composite Structures, (37)3, 259-272. doi:10.12989/scs.2020.37.3.259.
- [5] Zahrai, S. M., & Froozanfar, M. (2019). Improving Seismic Behavior of MRFs by U-shaped Hysteretic Damper Along Diagonal Brace. International Journal of Steel Structures, 19(2), 543–558. doi:10.1007/s13296-018-0139-2.
- [6] Javanmardi, A., Ghaedi, K., Ibrahim, Z., Huang, F., & Xu, P. (2020). Development of a new hexagonal honeycomb steel damper. Archives of Civil and Mechanical Engineering, 20(2), 1–19. doi:10.1007/s43452-020-00063-9.
- [7] Sahoo, D. R., Singhal, T., Taraithia, S. S., & Saini, A. (2015). Cyclic behavior of shear-and-flexural yielding metallic dampers. Journal of Constructional Steel Research, 114, 247–257. doi:10.1016/j.jcsr.2015.08.006.

- [8] Najari Varzaneh, M., & Hosseini, M. (2019). Cyclic Performance and Mechanical Characteristics of the Oval-shaped Damper. KSCE Journal of Civil Engineering, 23(11), 4747–4757. doi:10.1007/s12205-019-1382-6.
- [9] Mousavi, H., Sabbagh-Yazdi, S.-R. & Almohammad-albakkar, M., (2021). New mathematical formula for design viscous dampers in Internal and external Scissor-Jack braces, 6th International Conference on Interdisciplinary Researches in Civil Engineering, Architecture and Urban Management in 21st Century; 17 August, 2021, Tehran, Iran.
- [10] Zahrai, S. M., & Cheraghi, A. (2017). Improving cyclic behavior of multi-level pipe damper using infill or slit diaphragm inside inner pipe. Structural Engineering and Mechanics, 64(2), 195–204. doi:10.12989/sem.2017.64.2.195.
- [11] Kassem, M. M., Mohamed Nazri, F., & Noroozinejad Farsangi, E. (2020). The seismic vulnerability assessment methodologies: A state-of-the-art review. Ain Shams Engineering Journal, 11(4), 849–864. doi:10.1016/j.asej.2020.04.001.
- [12] Ghaedi, K. (2017). Earthquake Prediction. Earthquakes Tectonics, Hazard and Risk Mitigation, IntechOpen, London, United Kingdom. doi:10.5772/65511.
- [13] Javanmardi, A., Ibrahim, Z., Ghaedi, K., Benisi Ghadim, H., & Hanif, M. U. (2019). State-of-the-Art Review of Metallic Dampers: Testing, Development and Implementation. Archives of Computational Methods in Engineering, 27(2), 455–478. doi:10.1007/s11831-019-09329-9.
- [14] Boardman, P. R., Wood, B. J., & Carr, A. J. (1983). Union House a Cross Braced Structure With Energy Dissipators. Bulletin of the New Zealand National Society for Earthquake Engineering, 16(2), 83–97.
- [15] Martinez-Romero, E. (1993). Experiences on the use of supplementary energy dissipators on building structures. Earthquake Spectra, 9(3), 581–625. doi:10.1193/1.1585731.
- [16] Perry, C. L., Fierro, E. A., Sedarat, H., & Scholl, R. E. (1993). Seismic upgrade in San Francisco using energy dissipation devices. Earthquake Spectra, 9(3), 559–579. doi:10.1193/1.1585730.
- [17] Housner, G. W., Bergman, L. A., Caughey, T. K., Chassiakos, A. G., Claus, R. O., Masri, S. F., Skelton, R. E., Soong, T. T., Spencer, B. F., & Yao, J. T. P. (1997). Structural Control: Past, Present, and Future. Journal of Engineering Mechanics, 123(9), 897–971. doi:10.1061/(asce)0733-9399(1997)123:9(897).
- [18] Soong, T. T., & Spencer, B. F. (2002). Supplemental energy dissipation: State-of-the-art and state-of-the-practice. Engineering Structures, 24(3), 243–259. doi:10.1016/S0141-0296(01)00092-X.
- [19] Symans, M. D., Charney, F. A., Whittaker, A. S., Constantinou, M. C., Kircher, C. A., Johnson, M. W., & McNamara, R. J. (2008). Energy Dissipation Systems for Seismic Applications: Current Practice and Recent Developments. Journal of Structural Engineering, 134(1), 3–21. doi:10.1061/(asce)0733-9445(2008)134:1(3).
- [20] Saaed, T. E., Nikolakopoulos, G., Jonasson, J. E., & Hedlund, H. (2015). A state-of-the-art review of structural control systems. JVC/Journal of Vibration and Control, 21(5), 919–937. doi:10.1177/1077546313478294.
- [21] Li, H., & Huo, L. (2010). Advances in Structural Control in Civil Engineering in China. Mathematical Problems in Engineering, 2010, 1–23. doi:10.1155/2010/936081.
- [22] Ghaedi, K., Ibrahim, Z., Adeli, H., & Javanmardi, A. (2017). Invited review: Recent developments in vibration control of building and bridge structures. Journal of Vibroengineering, 19(5), 3564–3580. doi:10.21595/jve.2017.18900.
- [23] Korkmaz, S. (2011). A review of active structural control: Challenges for engineering informatics. Computers and Structures, 89(23–24), 2113–2132. doi:10.1016/j.compstruc.2011.07.010.
- [24] Dargush, G. F., & Sant, R. S. (2005). Evolutionary aseismic design and retrofit of structures with passive energy dissipation. Earthquake Engineering & Structural Dynamics, 34(13), 1601–1626. doi:10.1002/eqe.497.
- [25] Rahimi, F., Aghayari, R., & Samali, B. (2020). Application of tuned mass dampers for structural vibration control: A state-ofthe-art review. Civil Engineering Journal, 6(8), 1622–1651. doi:10.28991/cej-2020-03091571.
- [26] Fisco, N. R., & Adeli, H. (2011). Smart structures: Part I—Active and semi-active control. Scientia Iranica, 18(3), 275–284. doi:10.1016/j.scient.2011.05.034.
- [27] Fisco, N. R., & Adeli, H. (2011). Smart structures: Part II Hybrid control systems and control strategies. Scientia Iranica, 18(3), 285–295. doi:10.1016/j.scient.2011.05.035.
- [28] Behnamfar, F., & Almohammad-albakkar, M. (2023). Development of Steel Yielding Seismic Dampers Used to Improve Seismic Performance of Structures: A Comprehensive Review. International Journal of Engineering, 36(4), 746–775. doi:10.5829/ije.2023.36.04a.13.
- [29] Bakhshinezhad, S., & Mohebbi, M. (2020). Multi-objective optimal design of semi-active fluid viscous dampers for nonlinear structures using NSGA-II. Structures, 24, 678–689. doi:10.1016/j.istruc.2020.02.004.

- [30] Ghabussi, A., Asgari Marnani, J., & Rohanimanesh, M. S. (2021). Seismic performance assessment of a novel ductile steel braced frame equipped with steel curved damper. Structures, 31, 87–97. doi:10.1016/j.istruc.2021.01.073.
- [31] Wang, W., Song, J. liang, Su, S. qing, Cai, H. li, & Zhang, R. fu. (2021). Experimental and numerical studies of an axial tensioncompression corrugated steel plate damper. Thin-Walled Structures, 163, 107498. doi:10.1016/j.tws.2021.107498.
- [32] Lin, K., Zhou, A., Liu, H., Liu, Y., & Huang, C. (2020). Shear thickening fluid damper and its application to vibration mitigation of stay cable. Structures, 26, 214–223. doi:10.1016/j.istruc.2020.04.018.
- [33] Ghanbari, A., Mousavi, H., Almohammad-albakkar, M., & Habibi, M. R. (2022). Seismic Responses of Multi-Storey Structures Equipped with Linear and Nonlinear Viscous Dampers: A Comparison Study. SSRN Electronic Journal, 1-22. doi:10.2139/ssrn.4253077.
- [34] Setiawan, A. F., Santoso, A. K., Darmawan, M. F., Adi, A. D., & Ismanti, S. (2023). Nonlinear Analysis for Investigating Seismic Performance of a Spun Pile-Column of Viaduct Structure. Civil Engineering Journal, 9(7), 1561-1578. doi:10.28991/CEJ-2023-09-07-02.
- [35] Silwal, B., Michael, R. J., & Ozbulut, O. E. (2015). A superelastic viscous damper for enhanced seismic performance of steel moment frames. Engineering Structures, 105, 152-164. doi:10.1016/j.engstruct.2015.10.005.
- [36] Mousavi, H., Sabbagh Yazdi, S. R., & Almohammad-Albakkar, M. (2022). A novel method for efficient design of frame structures equipped with nonlinear viscous dampers by using computational results of cylindrical friction damper. Australian Journal of Structural Engineering, 24(1), 50–66. doi:10.1080/13287982.2022.2088055.
- [37] Gjukaj, A., Salihu, F., Muriqi, A., & Cvetanovski, P. (2023). Numerical Behavior of Extended End-Plate Bolted Connection under Monotonic Loading. HighTech and Innovation Journal, 4(2), 294-308. doi:10.28991/HIJ-2023-04-02-04.
- [38] Zheng, J., Zhang, C., & Li, A. (2019). Experimental investigation on the mechanical properties of curved metallic plate dampers. Applied Sciences, 10(1), 269. doi:10.3390/app10010269.
- [39] Chen, S. J., & Chang, C. C. (2012). Experimental study of low yield point steel gusset plate connections. Thin-Walled Structures, 57, 62–69. doi:10.1016/j.tws.2012.03.014.
- [40] Milanchian, R., & Hosseini, M. (2020). Torsional response reduction of plan-asymmetric vertical seismic isolation by appropriate distribution of viscous and viscoelastic dampers. Structures, 27, 962–974. doi:10.1016/j.istruc.2020.07.009.
- [41] Xiao, Y., Zhou, Y., & Huang, Z. (2021). Efficient direct displacement-based seismic design approach for structures with viscoelastic dampers. Structures, 29, 1699–1708. doi:10.1016/j.istruc.2020.12.067.
- [42] Abdul Aziz, M., Muhtasim, S., & Ahammed, R. (2022). State-of-the-art recent developments of large magnetorheological (MR) dampers: a review. Korea Australia Rheology Journal, 34(2), 105–136. doi:10.1007/s13367-022-00021-2.
- [43] Saravanan, M., Goswami, R., & Palani, G. S. (2018). Replaceable Fuses in Earthquake Resistant Steel Structures: A Review. International Journal of Steel Structures, 18(3), 868–879. doi:10.1007/s13296-018-0035-9.
- [44] Chen, S. J., & Jhang, C. (2006). Cyclic behavior of low yield point steel shear walls. Thin-Walled Structures, 44(7), 730–738. doi:10.1016/j.tws.2006.08.002.
- [45] Ayyash, A. jbury N. A., & Hejazi, F. (2023). Development of hybrid performance-based optimization algorithm for structures equipped with vibration damper devices. Archives of Civil and Mechanical Engineering, 23(2), 123. doi:10.1007/s43452-023-00665-z.
- [46] Ayyash, N., & Hejazi, F. (2022). Development of hybrid optimization algorithm for structures furnished with seismic damper devices using the particle swarm optimization method and gravitational search algorithm. Earthquake Engineering and Engineering Vibration, 21(2), 455–474. doi:10.1007/s11803-022-2088-1.
- [47] Ahmadie Amiri, H., Najafabadi, E. P., & Estekanchi, H. E. (2018). Experimental and analytical study of Block Slit Damper. Journal of Constructional Steel Research, 141, 167–178. doi:10.1016/j.jcsr.2017.11.006.
- [48] Dargush, G., & Soong, T. (1997). Passive Energy Dissipation and Active Control. Handbook of Structural Engineering, Second Edition, 1-28.
- [49] Christopoulos, C., & Filiatrault, A. (2006). Principles of passive supplemental damping and seismic. IUSS Press, Pavia, Italy.
- [50] Fukumoto, T. (1989). A Study on steel Damper with Honeycomb Shaped-Openings. Annual Meeting (Summaries of AIJ), Architectural Institute of Japan, Tokyo, Japan.
- [51] Naoki, T. (1991). A study on Steel Plate D. with Honeycomb-s. Openings s. to Low Cycle F. AIJ-37, Architectural Institute of Japan, Tokyo, Japan.
- [52] Wada, A., Huang, Y. H., Yamada, T., Ono, Y., Sugiyama, S., Baba, M., & Miyabara, T. (1997). Actual size and real time speed tests for hysteretic steel damper. Proceedings of Stessa, 97, 778–785.

- [53] Amadeo, B. C., OH, S. H., & Akiyama, H. (1998). Ultimate Energy Absorption Capacity of Slit-Type Steel Plates Subjected To Shear Deformations. Journal of Structural and Construction Engineering (Transactions of AIJ), 63(503), 139–147. doi:10.3130/aijs.63.139_1.
- [54] Javidan, M. M., & Kim, J. (2024). An experimental study on a steel multi-slit damper for seismic retrofit of soft-first story structures. Steel and Composite Structures, 50(6), 721–734. doi:10.12989/scs.2024.50.6.721.
- [55] Srisuwan, T., & Yooprasertchai, E. (2024). Effects of unbonded prestressing steel tendons and slit dampers on the seismic behavior of precast concrete beam-column joints. Structures, 59. doi:10.1016/j.istruc.2023.105721.
- [56] Payawal, J. M. G., & Kim, D. K. (2023). Evaluation of the Seismic Performance of Single-Plate Metallic Slit Dampers Using Experimental and Numerical Data. Buildings, 13(9), 2188. doi:10.3390/buildings13092188.
- [57] Lee, C. H., Ju, Y. K., Min, J. K., Lho, S. H., & Kim, S. D. (2015). Non-uniform steel strip dampers subjected to cyclic loadings. Engineering Structures, 99, 192–204. doi:10.1016/j.engstruct.2015.04.052.
- [58] González-Sanz, G., Escolano-Margarit, D., & Benavent-Climent, A. (2020). A new stainless-steel tube-in-tube damper for seismic protection of structures. Applied Sciences, 10(4), 1410. doi:10.3390/app10041410.
- [59] Zheng, J., Li, A., & Guo, T. (2015). Analytical and experimental study on mild steel dampers with non-uniform vertical slits. Earthquake Engineering and Engineering Vibration, 14(1), 111–123. doi:10.1007/s11803-015-0010-9.
- [60] Oh, S. H. (1998). Seismic Design of Energy Dissipating Multi-Story Frame with Flexible-Stiff Mixed Type Connection, Ph.D. Thesis, Tokyo University, Japan.
- [61] Chan, R. W. K., & Albermani, F. (2008). Experimental study of steel slit damper for passive energy dissipation. Engineering Structures, 30(4), 1058–1066. doi:10.1016/j.engstruct.2007.07.005.
- [62] Saffari, H., Hedayat, A. A., & Nejad, M. P. (2013). Post-Northridge connections with slit dampers to enhance strength and ductility. Journal of Constructional Steel Research, 80, 138-152. doi:10.1016/j.jcsr.2012.09.023.
- [63] Oh, S. H., Kim, Y. J., & Ryu, H. S. (2009). Seismic performance of steel structures with slit dampers. Engineering Structures, 31(9), 1997–2008. doi:10.1016/j.engstruct.2009.03.003.
- [64] Karavasilis, T. L., Kerawala, S., & Hale, E. (2012). Hysteretic model for steel energy dissipation devices and evaluation of a minimal-damage seismic design approach for steel buildings. Journal of Constructional Steel Research, 70, 358–367. doi:10.1016/j.jcsr.2011.10.010.
- [65] Hedayat, A. A. (2015). Prediction of the force displacement capacity boundary of an unbuckled steel slit damper. Journal of Constructional Steel Research, 114, 30–50. doi:10.1016/j.jcsr.2015.07.003.
- [66] Askariani, S. S., Garivani, S., & Aghakouchak, A. A. (2020). Application of slit link beam in eccentrically braced frames. Journal of Constructional Steel Research, 170, 106094. doi:10.1016/j.jcsr.2020.106094.
- [67] Roeder, C. W., & Popov, E. P. (1978). Eccentrically Braced Steel Frames for Earthquakes. Journal of the Structural Division (ASCE), 104(3), 391–412. doi:10.1061/jsdeag.0004875.
- [68] Bastami, M., & Ahmady Jazany, R. (2018). Development of Eccentrically Interconnected Braced Frame (EIC-BF) for seismic regions. Thin-Walled Structures, 131, 451–463. doi:10.1016/j.tws.2018.07.021.
- [69] Johnson S.M. (2005). Improved seismic performance of special concentrically braced frames. Master Thesis, University of Washington., Seattle, United States.
- [70] Metelli, G. (2013). Theoretical and experimental study on the cyclic behaviour of X braced steel frames. Engineering Structures, 46, 763–773. doi:10.1016/j.engstruct.2012.08.021.
- [71] Jazany, R. A., Hajirasouliha, I., & Farshchi, H. (2013). Influence of masonry infill on the seismic performance of concentrically braced frames. Journal of Constructional Steel Research, 88, 150–163. doi:10.1016/j.jcsr.2013.05.009.
- [72] Tremblay, R. (2001). Seismic behavior and design of concentrically braced steel frames. Engineering Journal, 38(3), 148–166. doi:10.62913/engj.v38i3.761.
- [73] Tremblay, R. (2002). Inelastic seismic response of steel bracing members. Journal of Constructional Steel Research, 58(5–8), 665–701. doi:10.1016/S0143-974X(01)00104-3.
- [74] Lumpkin, E. J., Hsiao, P. C., Roeder, C. W., Lehman, D. E., Tsai, C. Y., Wu, A. C., Wei, C. Y., & Tsai, K. C. (2012). Investigation of the seismic response of three-story special concentrically braced frames. Journal of Constructional Steel Research, 77, 131–144. doi:10.1016/j.jcsr.2012.04.003.
- [75] Karzad, A. S., Al-Sadoon, Z. A., Sagheer, A., & AlHamaydeh, M. (2022). Experimental and Nonlinear Finite Element Analysis Data for an Innovative Buckling Restrained Bracing System to Rehabilitate Seismically Deficient Structures. Data, 7(12), 171. doi:10.3390/data7120171.

- [76] Rafi, M. M., Lodi, S. H., Al-Sadoon, Z. A., Saatcioglu, M., & Palermo, D. (2022). Experimental Investigation of Dynamic Behavior of RC Frame Strengthened with Buckling-Restrained Bracing. Journal of Structural Engineering, 148(7), 4022076. doi:10.1061/(asce)st.1943-541x.0003371.
- [77] Rafi, M. M., Lodi, S. H., Al-Sadoon, Z. A., Saatcioglu, M., & Palermo, D. (2021). Shake-table testing of deficient reinforced concrete frame retrofitted with buckling restrained brace. ACI Structural Journal, 118(3), 161–173. doi:10.14359/51729351.
- [78] Al-Sadoon, Z. A., Saboor Karzad, A., Sagheer, A., & AlHamaydeh, M. (2022). Replaceable fuse buckling-restrained brace (BRB): Experimental cyclic qualification testing and NLFEA modeling. Structures, 39, 997–1015. doi:10.1016/j.istruc.2022.03.081.
- [79] Al-Sadoon, Z. A., Saatcioglu, M., & Palermo, D. (2020). New Buckling-Restrained Brace for Seismically Deficient Reinforced Concrete Frames. Journal of Structural Engineering, 146(6), 4020082. doi:10.1061/(asce)st.1943-541x.0002439.
- [80] Bastami, M., & Ahmady Jazany, R. (2019). Development of centrically fused braced frame (CFBF) for seismic regions. Soil Dynamics and Earthquake Engineering, 127, 105856. doi:10.1016/j.soildyn.2019.105856.
- [81] Yoo, J.-H., Roeder, C. W., & Lehman, D. E. (2008). Analytical Performance Simulation of Special Concentrically Braced Frames. Journal of Structural Engineering, 134(6), 881–889. doi:10.1061/(asce)0733-9445(2008)134:6(881).
- [82] Boostani, M., Rezaifar, O., & Gholhaki, M. (2018). Introduction and seismic performance investigation of the proposed lateral bracing system called "OGrid." Archives of Civil and Mechanical Engineering, 18(4), 1024–1041. doi:10.1016/j.acme.2018.02.003.
- [83] Lee, H. M., Oh, H. S., Huh, C., Oh, S. Y., Yoon, H. M., & Moon, S. T. (2002). Ultimate Energy Absorption Capacity of Steel Plate Slit Dampers Subjected to Shear Force. Steel Structures, 12(2), 71–79.
- [84] Benavent-Climent, A. (2010). A brace-type seismic damper based on yielding the walls of hollow structural sections. Engineering Structures, 32(4), 1113–1122. doi:10.1016/j.engstruct.2009.12.037.
- [85] Ghabraie, K., Chan, R., Huang, X., & Xie, Y. M. (2010). Shape optimization of metallic yielding devices for passive mitigation of seismic energy. Engineering Structures, 32(8), 2258–2267. doi:10.1016/j.engstruct.2010.03.028.
- [86] Aminzadeh, M., Kazemi, H. S., & Tavakkoli, S. M. (2020). A numerical study on optimum shape of steel slit dampers. Advances in Structural Engineering, 23(14), 2967–2981. doi:10.1177/1369433220927281.
- [87] Tagawa, H., Yamanishi, T., Takaki, A., & Chan, R. W. K. (2016). Cyclic behavior of seesaw energy dissipation system with steel slit dampers. Journal of Constructional Steel Research, 117, 24–34. doi:10.1016/j.jcsr.2015.09.014.
- [88] Katal Mohseni, P., Zahedi-khameneh, A., & Naeemifar, O. (2020). Study of the Effect of Geometric Parameters of Steel Block Slit Dampers on Energy Absorption. International Journal of Steel Structures, 20(3), 1069–1079. doi:10.1007/s13296-020-00343-3.
- [89] Kim, J. (2019). Development of seismic retrofit devices for building structures. International Journal of High-Rise Buildings, 8(3), 221–227. doi:10.21022/IJHRB.2019.8.3.221.
- [90] Guo, W., Ma, C., Yu, Y., Bu, D., & Zeng, C. (2020). Performance and optimum design of replaceable steel strips in an innovative metallic damper. Engineering Structures, 205, 110118. doi:10.1016/j.engstruct.2019.110118.
- [91] Askariani, S. S., & Garivani, S. (2020). Introducing and numerical study of a new brace-type slit damper. Structures, 27, 702– 717. doi:10.1016/j.istruc.2020.06.019.
- [92] Zhao, B., Lu, B., Zeng, X., & Gu, Q. (2021). Experimental and numerical study of hysteretic performance of new brace type damper. Journal of Constructional Steel Research, 183, 106717. doi:10.1016/j.jcsr.2021.106717.
- [93] Javidan, M. M., Nasab, M. S. E., & Kim, J. (2021). Full-scale tests of two-story RC frames retrofitted with steel plate multi-slit dampers. Steel and Composite Structures, 39(5), 645–664. doi:10.12989/scs.2021.39.5.645.
- [94] Benavent-Climent, A., Escolano-Margarit, D., Arcos-Espada, J., & Ponce-Parra, H. (2021). New metallic damper with multiphase behavior for seismic protection of structures. Metals, 11(2), 1–30. doi:10.3390/met11020183.
- [95] Almohammad-Albakkar, M., & Behnamfar, F. (2022). Numerical investigation of grooved gusset plate damper for using in cross-braced frames. Journal of Constructional Steel Research, 196, 107434. doi:10.1016/j.jcsr.2022.107434.
- [96] Almohammad-albakkar, M., Behnamfar, F., & Ataei, A. (2024). Experimental and numerical study of grooved gusset plate damper for cross-braced frames. Journal of Constructional Steel Research, 216, 108611. doi:10.1016/j.jcsr.2024.108611.
- [97] Almohammad-albakkar M. and Behnamfar F., (2023). Seismic Performance Assessment of Cross-Braced Steel Frame Equipped with New Damper, 13th International Congress on Civil Engineering, 9-11 May, 2023, Tehran, Iran.
- [98] Heyrani Moghaddam, S., & Shooshtari, A. (2023). Numerical and experimental investigation on seismic performance of proposed steel slit dampers. Journal of Constructional Steel Research, 200, 107646. doi:10.1016/j.jcsr.2022.107646.

- [99] Hui, C., Zhou, Z., Li, Y., Jiao, Y., & Hai, R. (2022). Quasi-static cyclic loading experiment and analysis of double-side slotted steel tube shear damper. Archives of Civil and Mechanical Engineering, 23(1), 45. doi:10.1007/s43452-022-00581-8.
- [100] Kang, H., Adane, M., Chun, S., & Kim, J. (2022). Development of a seismic retrofit system made of steel frame with vertical slits Development of a seismic retrofit system made of steel frame with vertical slits. Steel and Composite Structures, 44(2), 269–280.
- [101] Bae, J., Lee, C. H., Park, M., Alemayehu, R. W., Ryu, J., Kim, Y., & Ju, Y. K. (2020). Cyclic loading performance of radiuscut double coke-shaped strip dampers. Materials, 13(18), 3920. doi:10.3390/MA13183920.
- [102] Zhou, X., Tan, Y., Ke, K., Yam, M. C. H., Zhang, H., & Xu, J. (2023). An experimental and numerical study of brace-type long double C-section steel slit dampers. Journal of Building Engineering, 64, 105555. doi:10.1016/j.jobe.2022.105555.
- [103] He, L., Sun, X., Bu, H., & Tang, Z. (2022). Seismic Performance of MRSF Structures Damped with Steel Slit Shear Panels. International Journal of Structural Stability and Dynamics, 22(03n04), 2240013. doi:10.1142/s0219455422400132.
- [104] Ahmadie Amiri, H., Pournamazian Najafabadi, E., Esmailpur Estekanchi, H., & Ozbakkaloglu, T. (2020). Performance-based seismic design and assessment of low-rise steel special moment resisting frames with block slit dampers using endurance time method. Engineering Structures, 224, 110955. doi:10.1016/j.engstruct.2020.110955.
- [105] Oh, S. H., & Park, H. Y. (2022). Experimental study on seismic performance of steel slit damper under additional tensile load. Journal of Building Engineering, 50, 104110. doi:10.1016/j.jobe.2022.104110.
- [106] Lee, K. S., Lee, B. G., & Jung, J. S. (2022). Seismic Strengthening of R/C Buildings Retrofitted by New Window-Type System Using Non-Buckling Slit Dampers Examined via Pseudo-Dynamic Testing and Nonlinear Dynamic Analysis. Applied Sciences, 12(3), 1220. doi:10.3390/app12031220.